

\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{hyperref}

\graphicspath{{images/}}

\def\udcs{514.132} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{51M20, 51M25, 51M10} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{275}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
{
	\vbox{
		\noindent\logo
		\hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }
		
		\vspace{5mm}
		
		\begin{center}
			{\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
			{\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
			{\large Siberian Electronic Mathematical Reports} \\[1mm]
			{\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
			%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
			%     {\small Sobolev Institute of Mathematics SB RAS}
		\end{center}
		\vspace{-3mm}
		\noindent
		\begin{tabular}{c}
			\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
			\hline\hline
		\end{tabular}
		
		\vspace{1mm}
		{\flushleft\it Том 20, № 1 стр. 275--284 (2023) \hspace{55mm}{\rm\small УДК \udcs}} %Volume, pages in Russian,
		%filled by Editorial board
		\newline
		{\rm\small DOI~10.33048/semi.2023.20.022}\hphantom{aaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
	}%\vbox
}

\sloppy

\begin{document}
\begin{thebibliography}{19}
		
		\bibitem{AB13} N.V.~Abrosimov, G.A.~Baigonakova, 
		{\it Hyperbolic octahedron with $mmm$-symmetry}, Siberian Electronic Mathematical Reports, {\bf 10} (2013), 123--140. MR3039999, Zbl~1330.52014
		
		\bibitem{AM19} N.V.~Abrosimov, A.D.~Mednykh, 
		{\it Area and volume in non-Euclidean geometry}, Alberge, Vincent (ed.) et al., Eighteen essays in non-Euclidean geometry. Z\"{u}rich: European Mathematical Society (EMS). IRMA Lect. Math. Theor. Phys., {\bf 29} (2019), 151--189. MR3965122, Zbl~1416.51020
		
		\bibitem{AM14} N.V.~Abrosimov, A.D.~Mednykh, {\it Volumes of polytopes in constant curvature spaces} // Fields Inst. Commun., {\bf 70} (2014), 1--26. MR3329266, Zbl~1320.52013
		
		\bibitem{AV21} N.V.~Abrosimov, B.~Vuong,
		{\it Explicit volume formula for a hyperbolic tetrahedron in terms of edge lengths}, Journal of Knot Theory and its Ramifications, {\bf 30}:10 (2021), article number 2140007. MR4366412, Zbl~1482.52016
		
		\bibitem{Vinberg} D.V.~Alekseevskii, E.B.~Vinberg, A.S.~Solodovnikov, {\it Geometry of spaces of constant curvature}, in Geometry II: Spaces of Constant Curvature, Encyclopedia of Mathematical Sciences Vol.~29, Springer, 1993. MR1254932, Zbl~0787.53001
		
		\bibitem{B12} G.A.~Baigonakova,
		{\it Calculation of volumes of polyhedra in non-Euclidean geometry}, PhD thesis, Gorno-Altaisk, 2012.
		
		\bibitem{B69} S.~Bilinski, 
		{\it Zur Begr\"{u}ndung der elementaren Inhaltslehre in der hyperbolischen Ebene}, Math. Ann., {\bf 180} (1969), 256--268. MR0256259, Zbl~0159.21702
		
		\bibitem{Chokim} Yu.~Cho, H.~Kim, 
		{\it On the volume formula for hyperbolic tetrahedra}, Discrete and Computational Geometry, {\bf 22} (1999), 347--366. MR1706606, Zbl~0952.51013
		
		\bibitem{Derevnin} D.A.~Derevnin, A.D.~Mednykh, 
		{\it A formula for the volume of hyperbolic tetrahedron}, Russian Mathematical Surveys, {\bf 60}:2 (2005), 346--348.  MR2152953, Zbl~1085.51503
		
		\bibitem{Fenchel} W.~Fenchel,
		{\it Elementary geometry in hyperbolic space}, Walter de Gruyter, Berlin, New York, 1989. MR1004006, Zbl~0674.51001
		
		\bibitem{L50} F.~Lann\'{e}r, {\it On complexes with transitive groups of automorphisms}, Comm. Sem. Math. Univ. Lund, 1950. MR0042129, Zbl~0037.39802
		
		\bibitem{Lob}  N.I.~Lobatschefskij (N.I.~Lobachevsky),
		{\it Imagin\"are Geometrie und ihre Anwendung auf einige Integrale}, German translation by H.~Liebmann, Teubner, Leipzig, 1904.
		
		\bibitem{Mcconnell} B.~McConnell,
		{\it The laws of cosines for non-Euclidean tetrahedra}, preprint (2022), 
		\href{http://daylateanddollarshort.com/mathdocs/The-Laws-of-Cosines-for-Non-Euclidean-Tetrahedra.pdf}{http://daylateanddollarshort.com/mathdocs/The-Laws-of-Cosines-for-Non-Euclidean-}
		\href{http://daylateanddollarshort.com/mathdocs/The-Laws-of-Cosines-for-Non-Euclidean-Tetrahedra.pdf}{Tetrahedra.pdf}
		%\bibitem{Murush} J.~Murakami, A.~Ushijima,
	%	{\it A volume formula for hyperbolic tetrahedra in terms of edge lengths}, Journal of Geometry, {\bf 83}:1--2 (2005), 153--163.
		
		\bibitem{Muryano} J.~Murakami, M.~Yano, 
		{\it On the volume of a hyperbolic and spherical tetrahedron}, Communications in Analysis and Geometry, {\bf 13} (2005), 379--400. MR2154824, Zbl~1081.51012
		
		\bibitem{Nemoul} O.~Nemoul, N.~Mebarki,
		{\it Volume and boundary face area of a regular tetrahedron in a constant curvature space}, preprint (2018), arXiv:1803.10809 [physics.gen-ph]
		
		\bibitem{Sforza} G.~Sforza, 
		{\it Ricerche di estensionimetria negli spazi metrico-proiettivi}, Memorie Reale Accademia di Scienze, Lettere ed arti di Modena, {\bf III}, {\bf VIII} (Appendice) (1907), 21--66. JFM~38.0675.01
		
		\bibitem{Bolyai}  P.~St\"ackel, 
		{\it Geometrische Untersuchungen von Wolfgang Bolyai und Johann Bolyai}, I.–II., B.G. Teubner, Leipzig, 1913. JFM~44.0015.03
		
		\bibitem{Ushijima} A.~Ushijima, 
		{\it A volume formula for generalized hyperbolic tetrahedra}, in Non-Euclidean Geometries, Mathematics and Its Applications, eds. A.~Pr\'ekopa and E.~Moln\'ar, Vol.~581, Springer, 2006, pp. 249--265. MR2191251, Zbl~1096.52006
		
		
		%\bibitem{8} L.~Schl\"afli, 
		%{\it Theorie der vielfachen Kontinuit\"at}, Birkhauser, Basel, 1950.
		
	\end{thebibliography}

	\renewcommand{\refname}{References}
	\renewcommand{\proofname}{Proof.}
	\renewcommand{\figurename}{Fig.}
	\renewcommand{\tablename}{Table}
 

	\thispagestyle{empty}
	
	\title[The volume of a trirectangular hyperbolic tetrahedron]{The volume of a trirectangular\\ hyperbolic tetrahedron}
	\author{{N.V. ABROSIMOV, S.V. STEPANISHCHEV}}%
	\address{Nikolay Abrosimov \smallskip % Контактные данные всех авторов и место работы указываются только на английском языке
		\newline\hphantom{iii} Regional Scientific and Educational Mathematical Center, 
		\newline\hphantom{iii} Tomsk State University,
		\newline\hphantom{iii} pr. Lenina, 36,
		\newline\hphantom{iii} 634050, Tomsk, Russia\medskip
		\newline\hphantom{iii} Sobolev Institute of Mathematics,
		\newline\hphantom{iii} pr. Koptyuga, 4,
		\newline\hphantom{iii} 630090, Novosibirsk, Russia\smallskip}%
	\email{abrosimov@math.nsc.ru}%
	
	
	\address{Stepan Stepanishchev \smallskip % Контактные данные всех авторов и место работы указываются только на английском языке
		\newline\hphantom{iii} Novosibirsk State University,
		\newline\hphantom{iii} Pirogova str., 1,
		\newline\hphantom{iii} 630090, Novosibirsk, Russia\smallskip}%
	\email{step.stepanishchev@gmail.com}%
	
	\thanks{\sc Abrosimov, N.V., Stepanishchev, S.V.,
		The volume of a trirectangular hyperbolic tetrahedron}
	\thanks{\copyright \ 2022 Abrosimov N.V., Stepanishchev S.V}
	\thanks{\rm This work was supported by the Ministry of Science and Higher Education of Russia (agreement No. ~075-02-2023-943)}
	\thanks{\it Received December, 17, 2022, published March, 10,  2023.}%
	
	
	\semrtop \vspace{1cm}
	\maketitle {\small
		\begin{quote}
			\noindent{\sc Abstract. } We consider a three-parameter family of tetrahedra in the hyperbolic space, which three edges at one vertex are pairwise orthogonal. It is convenient to determine such tetrahedra by the lengths of these edges. We obtain relatively simple formulas for them expressing the volume and the surface area. This allows us to find normalized volume and investigate its asymptotics.\medskip
			
			\noindent{\bf Keywords:} hyperbolic volume, normalized volume, Poincar\'{e} upper half-space model, hyperbolic tetrahedron, trirectangular tetrahedron, infinite cone.
		\end{quote}
	}
	
	
	\section{Introduction}
	Finding the volume of a hyperbolic polyhedron is a very old and difficult problem. The volume of a biorthogonal hyperbolic tetrahedron (so called {\em orthoscheme}) was found by N.~Lobachevsky \cite{Lob} and J.~Bolyai \cite{Bolyai} indepentently. In 1907, G.~Sforza \cite{Sforza} proposed a formula expressing the volume of an arbitrary hyperbolic tetrahedron in terms of dihedral angles. Before presenting it, we introduce some definitions.
	
	A {\em hyperbolic tetrahedron} $T$ is a convex hull of four points in the hyperbolic space $\mathbb{H}^3$. These points are called {\em vertices} of $T$. Let us denote them by numbers $1, 2, 3$ and $4$ (see Fig.\ref{fig1}). Then denote by $\ell_{ij}$ the length of the edge connecting $i$-th and $j$-th vertices. We put $\theta_{ij}$ for the dihedral angle along the corresponding edge.
	\begin{figure}[th]
		\centering
		\includegraphics[width=0.4\linewidth]{fig1.eps}
		\caption{Hyperbolic tetrahedron $T$}\label{fig1}
	\end{figure}
	
	A {\em Gram matrix} $G(T)$ of tetrahedron $T$ is defined as 
	$$
	G(T)=
	%\langle-\cos\theta_{ij}\rangle_{i,j=1,2,3,4}=
	\left(%
	\begin{array}{cccc}
	1 & -\cos\theta_{12} & -\cos\theta_{13} & -\cos\theta_{14} \\
	-\cos\theta_{12} & 1 & -\cos\theta_{23} & -\cos\theta_{24} \\
	-\cos\theta_{13} & -\cos\theta_{23} & 1 & -\cos\theta_{34} \\
	-\cos\theta_{14} & -\cos\theta_{24} & -\cos\theta_{34} & 1 \\
	\end{array}%
	\right).
	$$ 
	%we assume here that $-\cos\theta_{ii}=1$.
	
	An {\em Edge matrix} $E(T)$ is formed by hyperbolic cosines of the edge lengths and defined as follows
	$$E(T)=
	%\langle\cosh\ell_{ij}\rangle_{i,j=1,2,3,4}=
	\left(%
	\begin{array}{cccc}
	1 & \cosh\ell_{12} & \cosh\ell_{13} & \cosh\ell_{14} \\
	\cosh\ell_{12} & 1 & \cosh\ell_{23} & \cosh\ell_{24} \\
	\cosh\ell_{13} & \cosh\ell_{23} & 1 & \cosh\ell_{34} \\
	\cosh\ell_{14} & \cosh\ell_{24} & \cosh\ell_{34} & 1 \\
	\end{array}%
	\right).
	$$ 
	%where $\ell_{ii}=0$ and $\cosh\ell_{ii}=1$.
	
	It is known that a hyperbolic tetrahedron $T$ can be uniquely determined up to isometry either by the Gram matrix $G(T)$ or the edge matrix $E(T)$ (see, e.g., \cite{Vinberg}).
	
	\begin{theorem}[G. Sforza, 1907]\label{t1}
		Let $T$ be a compact hyperbolic tetrahedron given by its Gram matrix $G$. We assume that all the dihedral angles are fixed exept $\theta_{34}$ which is formal variable. Then the volume $V=V(T)$ is given by the formula
		\begin{equation*}
		V=\frac{1}{4}\int\limits_{t_0}^{\theta_{34}}\log\frac{c_{34}(t)-\sqrt{-\det G(t)}\sin t}{c_{34}(t)+\sqrt{-\det G(t)}\sin t}dt,
		\end{equation*}
		where $t_0$ is a suitable root of the equation $\det G(t) = 0$, $c_{34}$ is
		$(3, 4)$-cofactor of the matrix $G$, and $c_{34}(t), \det G(t)$ are functions in one variable $\theta_{34}$ denoted by $t$.
	\end{theorem}

	A new proof of the classical Sforza's formula and its version for the spherical tetrahedron can be found in the paper by the first author and A.~Mednykh \cite{AM14}.
	An analog of the Sforza's formula for the volume of an arbitrary compact hyperbolic tetrahedron but in terms of edge lengths instead of dihedral angles was given in the recent paper by the first author and B.~Vuong \cite{AV21}.
	
%	\begin{theorem}[N. Abrosimov, B.~Vuong, 2021]
%		Let $T$ be a compact hyperbolic tetrahedron given by its edge matrix $E$ and $c_{ij}=(-1)^{i+j}E_{ij}$ is $ij$-cofactor of $E$. We assume that all the edge lengths are fixed exept $\ell_{34}$ which varies. Then the volume $V=V(T)$ is given by the formula 
%		\begin{multline*}
%		V=\frac12\int\limits_{\ell_1}^{\ell_{34}}\Bigl[\frac{-t}{\sqrt{-\Delta^{3}}}\Bigl(\frac{c_{14}(c_{11}c_{23}-c_{12}c_{13})}{c_{11}}+\frac{c_{24}(c_{13}c_{22}-c_{12}c_{23})}{c_{22}}\Bigr)-\frac{\sinh t}{\sqrt{-\Delta}}\cdot\\
%		\Bigl(\frac{\ell_{24}\sinh\ell_{24}c_{14}+\ell_{14}\sinh\ell_{23}c_{13}}{c_{11}}+\frac{\ell_{13}\sinh\ell_{13}c_{23}+\ell_{23}\sinh\ell_{14}c_{24}}{c_{22}}+\ell_{12}\sinh\ell_{12}\Bigr)\Bigr]dt,
%		\end{multline*}
%		where cofactors $c_{ij}$ and edge matrix determinant $\Delta=\det E$ are functions in one variable $\ell_{34}$ denoted by $t$.
%	\end{theorem}
	
	There are also three more known formulas expressing the volume of an arbitrary hyperbolic tetrahedron in terms of dihedral angles. In 1999 solution of the problem was introduced by Yu.~Cho and H.~Kim \cite{Chokim}. Then another formula was given by J.~Murakami and M.~Yano \cite{Muryano}. A.~Ushijima \cite{Ushijima} found a geometrical proof for this result which also covers the case of truncated tetrahedron. In 2005 D.~Derevnin and A.~Mednykh \cite{Derevnin} proposed a closed integral formula.
	
	Known formulas for the volume of a general hyperbolic tetrahedron are rather complicated and not so convenient for applications including volume calculations for more complex polyhedra in $\mathbb{H}^3$. The goal of the present paper is to provide with a new comparatively simple formula for sufficiently large subfamily of hyperbolic tetrahedra. 
	
	A {\em trirectangular tetrahedron} is a tetrahedron where all three face angles at one vertex are right angles. That vertex is called the {\em right angle vertex} of the trirectangular tetrahedron and the face opposite it is called the {\em base}. 
	The three edges that meet at the right angle are called the {\em legs} of a trirectangular tetrahedron. 
	
	Corresponding $3$-parameter family of hyperbolic tetrahedra was considered in the PhD thesis by G.~Baigonakova \cite{B12}, where trigonometrical relations between the dihedral angles and edge lengths were established. 
	
	If we consider reflections in three pairwice orthogonal faces of a trirectangular tetrahedron $T$ then we obtain a hyperbolic octahedron with so called $mmm$-symmetry. Such an octahedron was investigated in the paper by the first author and G.~Baigonakova \cite{AB13}, where the volume was find in terms of its dihedral angles.
	
	A {\em Coxeter polyhedron} is a polyhedron with all dihedral angles of the form $\pi/n$. F.~Lann\'{e}r \cite{L50} proved that there are exactly $9$ compact Coxeter tetrahedra in $\mathbb{H}^3$. One of them $T(\theta_{12},\theta_{13},\theta_{23},\theta_{34},\theta_{24},\theta_{14})=T(\pi/2,\pi/2,\pi/3,\pi/3,\pi/5,\pi/2)$ is trirectangular tetrahedron. The remaining $8$ Coxeter tetrahedra are ortoschemes. Both ortoschemes and trirectangular tetrahedra, not necessarily Coxeter ones, can be used for triangulations of more complex polyhedra in $\mathbb{H}^3$.
	
	In the present paper we consider a trirectangular tetrahedron in the hyperbolic spase given by the lengths of its legs $l_1, l_2, l_3$. We obtain relatively simple formulas for it expressing the volume and the surface area. This allows us to find normalized volume and investigate its asymptotics.
	
	Most of known results for the volume of a hyperbolic tetrahedron are based on the classical Schl\"{a}fli differential equation (see, e.g., \cite{Vinberg}, Ch.~7, Sect.~2.2).
	\begin{gather*}
	-dV = \frac{1}{2}\sum_{ij}\ell_{ij}\,d\theta_{ij}, 
	\end{gather*}
	where the sum is taken over all edges.
	
	In the present work instead of using Schl\"{a}fli equation, we integrate the volume element in $\mathbb{H}^3$ and use Fubini's theorem. 
	
	
	
	\section{Volume formula}
	Consider a Poincar\'{e} model of $\mathbb{H}^3$. That is the upper half-space $\mathbb{R}^3_+ = \left\{ (x, y, z) \in \mathbb{R}^3\, |\, z > 0\right\}$  endowed with the metric
	\begin{align*}
	d s^2 = \frac{d x^2 + d y^2 + d z^2}{z^2}.
	\end{align*}
	
	The volume element in this model has a form (see \cite{Vinberg}, Ch.~7, Sect.~2.3)
	\begin{align}\label{dvol}
	d V = \frac{\,dx\,dy\,dz}{z^3}.
	\end{align}
	
	
		Let $f(x, y)$ be a real function defined on domain $D \subset \mathbb{R}^2$. An {\em infinite cone} $C$ over the graph of function $f$ is the set formed by vertical lines starting on this graph
		\begin{equation*}
		C = \left\{ (x, y, z) \in \mathbb{R}^3_+\; |\; z \geq f(x, y),\, (x, y) \in D\right\}.
		\end{equation*}

	Using (\ref{dvol}) and Fubini's theorem we get the volume of infinite cone $C$ over  the graph of function $f(x,y)$
	\begin{gather}\label{vol}
	V(C) = \iiint\limits_C\frac{\,dx\,dy\,dz}{z^3} = \iint\limits_D\int\limits_{f(x, y)}^\infty\frac{\,dx\,dy\,dz}{z^3} = \frac{1}{2}\iint\limits_D\frac{dx\,dy}{(f(x, y))^2}.
	\end{gather}
	
	Consider a trirectangular tetrahedron $T=T(\ell_1, \ell_2, \ell_3)$ given by the lengths of its legs (i.e. pairwise ortogonal edges). Let us enumerate the vertices $1, \dots, 4$ in such a way that $\ell_{13} = \ell_1,$ $\ell_{14} = \ell_2,$ $\ell_{12} = \ell_3$.  There exist an isometry of $\mathbb{H}^3$ which maps vertex $1$ to  point $(0,\,0,\,1)\in\mathbb{R}^3_+$, guides $\ell_3$ along the axis $0z$, and put vertices $3$ and $4$ in coordinate planes $0xz, 0yz$ correspondingly (Fig.~\ref{fig2}). We refer this configuration of tetrahedron $T$ in $\mathbb{R}^3_+$ as {\em standard position}. Further we assume without loss of generality that $T$ is in this position.
	\begin{figure}[th]
		\centering
		\includegraphics[width=0.5\linewidth]{fig2.eps}
		%    \hspace{50pt}
		%    \vspace{-25pt} width=0.5\linewidth
		\caption{Hyperbolic tetrahedron $T$ in standart position\label{fig2}}
	\end{figure}
	\begin{theorem} \label{OurTheorem}
		Let $T = T(\ell_1, \ell_2, \ell_3)$ be a compact hyperbolic tetrahedron defined by lengths of pairwise ortogonal edges. Then the volume $V = V(T)$ is given by the formula
		\begin{eqnarray}
		\nonumber
		V = \frac{1}{2}\int\limits_0^{\tanh{\ell_1}}\int\limits_0^\frac{\tanh{\ell_2}(\tanh{\ell_1} - x)}{\tanh{\ell_1}}\left[\frac{1}{1 - x^2 - y^2} + \frac{1}{x(x - 2x_0) + y(y - 2y_0) - e^{2\ell_3}}\right]dx\,dy,
		\end{eqnarray}
		where\, $x_0 = \dfrac{1-e^{2\,\ell_3}}{2\tanh{\ell_1}}$ \,and\; $y_0 = \dfrac{1-e^{2\,\ell_3}}{2\tanh{\ell_2}}$.
	\end{theorem}
	\begin{proof}
		Let $Q$ and $P$ be the lower and upper bases of $T$. Denote by $V_Q$ and $V_P$ the volumes of infinite cones over them. Then the required volume $V = V_Q - V_P$.
		We use formula (\ref{vol}) in order to calculate $V_Q$ and $V_P$. To do this we define the functions $f_Q(x, y)$ and $f_P(x, y)$ which determine hyperplanes $Q, P$ then find the limits of integration.
		
		Face $Q$ is the part of Euclidean unit sphere $x^2 + y^2 + z^2 = 1$, since it is spanned by the edges $\ell_1$ and $\ell_2$ orthogonal to $Oz$. Then
			\begin{eqnarray*}
				f_Q^2(x, y) = 1 - x^2 - y^2.
			\end{eqnarray*}
		
			
        To find $f_P(x, y)$ we define the coordinates of vertices $2, 3$ and $4$.
		
        Let the vertex $2$ has coordinates $(x_2,\, y_2,\, z_2)$. Since it lies on axis $Oz$ we have $x_2 = y_2 = 0$. In addition, we know that the distance between $1$ and $2$ is $\rho(1, 2) = \ell_3$. On the other hand, (see \cite{Fenchel}, Ch.~III, Sect.~III.4) $$\rho(1, 2) = \int\limits_1^{z_2}\frac{dz}{z} = \ln{z_2}.$$
		Therefore, $z_2 = e^{\ell_3}$ and 2 has coordinates $(0,\,0,\,e^{\ell_3})$.
				
        Let $\varphi_1$ be the angle between axis $0z$ and radius-vector of vertex $3$ (Fig. \ref{fig3}). This vertex lies on the edge $\ell_1$ which is the part of unit semi-circle. So $3$ has coordinates $(\sin{\varphi_1},\, 0,\, \cos{\varphi_1})$.
				\begin{figure}[h]
					\centering
					\includegraphics[width=0.5\linewidth]{fig3.eps}
					\vspace{-30pt}
					\caption{Angle $\varphi_1$ between axis $0z$ and radius-vector of vertex $4$\label{fig3}}
					%\label{img:firstver} 
				\end{figure}
				Hyperbolic length of the arc connecting vertices $1$ and $3$ is $\ell_1$. At the same time, $\ell_1$ and $\varphi_1$ are related as follows (\cite{Fenchel}, Ch.~III, Sect.~III.4) 
				\begin{eqnarray*}\label{chcos}
					\cosh{\ell_1} = \frac{1}{\cos{\varphi_1}},
				\end{eqnarray*}
				where we can also get $\tanh{\ell_1} = \sin{\varphi_1}$.
				Thereby, vertex $3$ has coordinates $\left(\tanh{\ell_1},\,0,\,\dfrac{1}{\cosh{\ell_1}}\right)$.
				
        For the $4$-th vertex we can get coordinates $\left(0,\,\tanh{\ell_2},\, \dfrac{1}{\cosh{\ell_2}}\right)$ in the same way.
			
		Let hyperbolic plane (i.e. Euclidean semi-sphere) that contains $P$ has a center $\left(x_0,\,y_0,\,z_0\right)$. Take into consideration that $z_0 = 0$ as a center of any hyperbolic plane lies on the absolute. In general we have
			\begin{eqnarray*}\label{sphere}
				(x - x_0)^2 + (y - y_0)^2 + z^2 = R^2.
			\end{eqnarray*}
			Substituting the coordinates of vertices $2,\,3,\,4$ in this equation we get the system of three equations with variables $x_0$, $y_0$, $R$
			\begin{eqnarray*}\label{cases}
				\begin{cases}
					\left(\tanh{\ell_1} - x_0\right)^2 + y_0^2 + \left(\dfrac{1}{\cosh{\ell_1}}\right)^2 = R^2, \\        
					x_0^2 + \left(\tanh{\ell_2} - y_0\right)^2 + \left(\dfrac{1}{\cosh{\ell_2}}\right)^2 = R^2, \\
					x_0^2 + y_0^2 + e^{2\ell_3} = R^2.
				\end{cases}
			\end{eqnarray*}
		
    Solving that we finally obtain 
    $$
	f_P^2(x, y) = -x(x - 2x_0) - y(y - 2y_0) + e^{2l_3},
	$$
	where $x_0$ and $y_0$ are given by relations \;$x_0 = \dfrac{1-e^{2\,\ell_3}}{2\tanh{\ell_1}}$ \,and\; $y_0 = \dfrac{1-e^{2\,\ell_3}}{2\tanh{\ell_2}}$.
			
    Integration in formula (\ref{vol}) is taken over the domain of function $f(x, y)$. In our case $Q$ and $P$ are projected on the absolute into triangle $D$, that is the domains of integration for $V_Q$ and $V_P$ coincides. Thus, we can rewrite difference of integrals as integral of difference.
			\begin{figure}[h]
				\centering
				\includegraphics[width=0.5\linewidth]{fig4.eps}
				\vspace{-30pt}
				\caption{Projection of faces $P$ and $Q$ on the absolute\label{fig4}}
				%\label{img:triangle} 
			\end{figure}
			
   Consider triangle $D$ (Fig.~\ref{fig4}). Its vertices are $\left(0,\, 0\right)$, $\left(\tanh{\ell_1},\, 0\right)$ and $\left(0,\, \tanh{\ell_2}\right)$. We find equation of the straight line passing through the last two points
			\begin{equation*}
				y = \frac{\tanh{\ell_2}(\tanh{\ell_1} - x)}{\tanh{\ell_1}}.
			\end{equation*}
	Therefore, while $x$ varies within $0$ and $\tanh{\ell_1}$ the second coordinate $y$ runs from $0$ to $\dfrac{\tanh{\ell_2}(\tanh{\ell_1} - x)}{\tanh{\ell_1}}$.
		
	Now we can apply (\ref{vol}) with functions $f_Q(x, y)$, $f_P(x, y)$ and prescribed integration limits to find $V_Q$ and $V_P$. Thus we get the fromula from statement of the theorem using the fact that $V = V_Q - V_P$.
	\end{proof}
	
	\section{Formula verification}
	
	To check the formula we have obtained, we use Sforza's formula (see Theorem~\ref{t1}). We take $\ell_1 = \ell_2 = \ell_3 = \ell$.

	%\clearpage
	\begin{table}[h]
	     \centering
	     \begin{tabular}{c|c|c|c}
	     	  $\ell$ & Theorem~1  & Theorem~2 \\[3pt]
	    	0.25 & 0.0025398399\ldots  & 0.0025398399\ldots \\[3pt]
	         0.5  & 0.0188499596\ldots  & 0.0188499596\ldots \\[3pt]
	          1    & 0.1124224663\ldots  & 0.1124224667\ldots \\[3pt]
	          2    & 0.3463865845\ldots  & 0.3463863000\ldots \\[3pt]
	          4    & 0.4538079351\ldots  & 0.4538079400\ldots \\[3pt]
	          8    & 0.4579801973\ldots  & 0.4579810839\ldots 
	     \end{tabular}
     	\bigskip
	     \caption{Volume of a tetrahedron $T$, calculated by different formulas}\label{tab:tab1}
	\end{table}\vspace{-15pt}
	
	We note that for large values of $\ell$, the integral in Sforza's formula accumulates a calculation error while the formula in Theorem~2 works well. As $\ell\rightarrow\infty$, the volume tends to $1/8$ of the volume of regular ideal octahedron, that is 
	\begin{equation*}
	\lim\limits_{\ell\to\infty} V(T) = \dfrac{1}{2}\,G = 0,457982797\ldots,
	\end{equation*}
	 where $G = \displaystyle\sum\limits_{n=0}^\infty\frac{(-1)^n}{(2n+1)^2} = 0.9159655941\ldots$ is Catalan's constant.
	 
\section{Normalized volume}
	In the work by O.~Nemoul and N.~Mebarki \cite{Nemoul} the normalized volume of a compact regular hyperbolic tetrahedron is given. Let us find it for $3$-parameter family of tetrahedra under consideration. Following \cite{Nemoul} we define {\em normalized volume} as $\nu(T)=\dfrac{V}{S^{3/2}}$, where $V$ is the volume of $T$ and $S$ is its surface area.
	
	Let $\mathbf{X}, \mathbf{Y}, \mathbf{Z}$ be the areas of rectangular faces 1--2--3, \,1--2--4, \,1--3--4 correspondingly and $\mathbf{W}$ is the area of 2--3--4 (Fig.~\ref{fig:area}).
	\begin{figure}[h]
		\centering
		\includegraphics[scale=0.79]{fig5.eps}
		%\vspace{-25pt}
		\caption{Areas of the faces of a tetrahedron $T$}\label{fig:area}
	\end{figure}

	One of the analogs of Heron's formula for the area of a hyperbolic triangle is known as Bilinsky formula \cite{B69} (see also \cite{AM19}). 
	\begin{eqnarray}\label{heron}
	\cos{\frac{\mathbf{X}}{2}} = \frac{\cosh{\ell_1} + \cosh{\ell_3} + \cosh{h} + 1}{4\cosh{\frac{\ell_1}2}\cosh{\frac{\ell_3}2}\cosh{\frac{h}2}},
	\end{eqnarray}
	where $h$ is hypotenuse in the triangle 1--2--3. By Pythagorean theorem
	\begin{eqnarray}\label{hip}
	\cosh h = \cosh{l_1}\cosh{l_3}.
	\end{eqnarray}
	We substitute (\ref{hip}) in (\ref{heron}) and use equality $1 + \cosh{\ell} = 2\cosh^2{\frac{\ell}{2}}$ to find
	\begin{equation*}
	\cos{\frac{\mathbf{X}}{2}} = \frac{\cosh{\frac{\ell_1}2}\cosh{\frac{\ell_3}2}}{\cosh{\frac{h}2}},  
	\end{equation*}
	and equivalently
	\begin{equation*}
	\sin{\frac{\mathbf{X}}{2}} = \frac{\sinh{\frac{\ell_1}2}\sinh{\frac{\ell_3}2}}{\cosh{\frac{h}2}}.
	\end{equation*}
	Using these relations for $\mathbf{X}$, $\mathbf{Y}$, $\mathbf{Z}$ we get the areas of rectangular faces
	\begin{align*}
	\mathbf{X} &= 2\arctan\left(\tanh{\frac{\ell_1}2}\tanh{\frac{\ell_3}2}\right), \\
	\mathbf{Y} &= 2\arctan\left(\tanh{\frac{\ell_2}2}\tanh{\frac{\ell_3}2}\right), \\
	\mathbf{Z} &= 2\arctan\left(\tanh{\frac{\ell_1}2}\tanh{\frac{\ell_2}2}\right).
	\end{align*}
	
	Then we use a hyperbolic version of De Gua's theorem for the faces of a tetrahedron given by B.~McConnell \cite{Mcconnell} to get the face area $\mathbf{W}=$
	\begin{equation*}
		 2\arccos{\frac{1 - \tanh^2{\frac{\ell_1}2}\tanh^2{\frac{\ell_2}2}\tanh^2{\frac{\ell_3}2}}{\sqrt{(1 + \tanh^2{\frac{\ell_1}2}\tanh^2{\frac{\ell_3}2})(1 + \tanh^2{\frac{\ell_2}2}\tanh^2{\frac{\ell_3}2})(1 + \tanh^2{\frac{\ell_1}2}\tanh^2{\frac{\ell_2}2})}}}.
	\end{equation*}
	Thus, the normalized volume $\nu(T) =\displaystyle\frac{V}{S^{3/2}}$ is calculated by Theorem~2 and $S=\mathbf{X} + \mathbf{Y} + \mathbf{Z} + \mathbf{W}$.
	
	To find the asymptotics of the normalized volume we take $\ell_1 = \ell_2 = \ell_3 = \ell$. As $\ell \to \infty$ the faces areas $\mathbf{X}$, $\mathbf{Y}$, and $\mathbf{Z}$ attend to $\displaystyle\frac{\pi}2$ as the areas of right triangles with two vertices at infinity. The face area $\mathbf{W}$ attends to $\pi$ as the area of a regular ideal hyperbolic triangle. Therefore, $\lim\limits_{\ell \to \infty} S = \displaystyle\frac{5 \pi}2$. Thus,
	$$\lim\limits_{\ell \to \infty} \nu = \lim\limits_{\ell \to \infty}\displaystyle\frac{V}{S^{3/2}} = \frac{\sqrt{2}\,G}{(5\pi)^{3/2}} = 0.0208071557\dots,$$
	where $G = 0.9159655941\ldots$ is Catalan's constant.
	
	Consider the behavior of the normalized volume $\nu(T)=\nu(\ell)$ for the case $\ell_1 = \ell_2 = \ell_3 = \ell$ (Fig.~\ref{fig:graph}). It monotonically decreases for $\ell\in(0,+\infty)$.
	\begin{figure}[h]
		\centering
		\includegraphics[scale=0.79]{fig6.eps}
		%\vspace{-25pt}
		\caption{Graph of the normalized volume of a tetrahedron $T$}\label{fig:graph}
	\end{figure}
	
	\bigskip
	
\end{document}
